Author's response to reviews

Title: Determination of the social related factors of suicide in Iran: A systematic review and meta-analysis

Authors:

Milad Nazarzadeh Mr (nazarzadeh_milad@yahoo.com)
Zeinab Bidel Miss (Zeinab.bidel@yahoo.com)
Erfan Ayubi Mr (Aubi65@gmail.com)
Asadollahi Khirollah Dr (masoud_1241@yahoo.co.uk)
Kourosh Sayehmiri Dr (kouresh_sayehmiri@yahoo.com)

Version: 6 Date: 13 November 2012

Author's response to reviews: see over
Dear Dr. Jimmar Dizon

We appreciate your consideration to our article. The grammatical and syntax error were corrected and a sophisticated English Professor cheked the entire article with scrutiny. Mean which, it is suited to your Journal format and all the points in your previous email were regarded. We hope our article will get the honor for publication in your journal.

Thank you for your attention. I look forward to hearing from you.

Yours sincerely,

Dr. Kourosh Sayehmiri

Address: Psychosocial Injuries Research Center, Ilam University of Medical Sciences, Ilam, Iran.

Tel: +989183410782, Fax:+988413342887

Assistance Professor of Biostatistics